Author's response to reviews

Title: Circulating levels of cell adhesion molecule L1 as a prognostic marker in gastrointestinal stromal tumor patients

Authors:

Hilke Zander (hilke.zander@zmnh.uni-hamburg.de)
Tamina Rawnaq (trawnaq@uke.de)
Max .Wedemeyer (m.vonwedemeyer@googlemail.com)
Michael Tachezy (mtachezy@uke.de)
Miriam Kunkel (mkunkel@uke.de)
Gerrit Wolters (ge.wolters@uke.de)
Max Bockhorn (m.bockhorn@uke.de)
Melitta Schachner (Schachner@Biology.Rutgers.Edu)
Jacob R Izbicki (izbicki@uke.de)
Jussuf T Kaifi (jkaifi@hmc.psu.edu)

Version: 3 Date: 6 April 2011

Author's response to reviews: see over
Dear Dr. Bednarek,

Please consider our revised manuscript

“Circulating levels of cell adhesion molecule L1 as a prognostic marker in gastrointestinal stromal tumor patients”.

We would like to thank you and the reviewers for your comments and suggestions which helped us to improve the manuscript. The manuscript was revised in accordance with the reviewer’s suggestions and hope that we adequately addressed the concerns. The specific change to the manuscript is highlighted in the revised version.

We hope the revised manuscript is now suitable for publication.

There has been no duplicate publication or submission elsewhere, all authors have read and approved the manuscript. Subject to acceptance, authors will transfer copyright to the Publisher. There is no ethical problem or conflict of interest.

Sincerely yours

Hilke Zander PhD
Corresponding author